Author's response to reviews

Title: Effect of prophylactic 360 degrees laser treatment for prevention of retinal detachment after phacovitrectomy.

Authors:

Takeshi Iwase (tsuyoshiwase@aol.com)
Brian C Oveson (peterdodemoiiyo@yahoo.com)
Young-Joon Jo (tonamimieko1@yahoo.co.jp)

Version: 3
Date: 26 November 2013

Author's response to reviews: see over
Dear Prof. Murray

We deeply appreciate your generous consideration and comments on our manuscript entitled “Effect of prophylactic 360 degrees laser treatment for prevention of retinal detachment after phacovitrectomy.” Here, we address the comments and explain how we modified the text accordingly.

Reviewer: Annal Meleth

Minor essential revisions:
A number of small grammatical errors remain which should be addressed in the process of editing for publication; primarily preposition usage.
Examples: in the "other complications" subsection of the results:
1. "Proportions of ERM" should be changed to "The proportion of ERM/The prevalence of ERM formation"
2. "Postoperative ERM was developed" should be changed to "Postoperative ERM developed"
3. "Two eyes had the posterior capsular rupture" should be changed to "Two eyes had rupture of the posterior capsule/Two eyes had posterior capsular rupture"

-------- We have corrected the grammatical errors, and altered the syntax to improve the stylistic aspects of the manuscript.

We would like to thank the reviewers for their comments and suggestions and we hope the reviewers will be satisfied with our revisions. We are looking forward to hearing from you.

P.S.) None of the authors have any financial or proprietary interests in any product
mentioned.

Sincerely yours,

Takeshi Iwase, MD,
Wilmer Eye Institute,
Johns Hopkins University School of Medicine,
846 Maumenee, 600 N. Wolfe Street,
Baltimore, MD, USA 21287-9277
tel: 410-955-4103,
fax: 410-614-6619;
e-mail: TsuyoshiIwase@aol.com